Author’s response to reviews

Title: Short-term Effects and Safety of an acute increase of intraocular pressure after Intravitreal Bevacizumab injection on Corneal Endothelial Cells

Authors:

jiwoong park (pjiwoong@hanmail.net)

myungwon lee (persica79@naver.com)

Version: 3 Date: 02 Oct 2017

Author’s response to reviews:

Editor Comments:

1. BioMed Central requests a trial registration number for manuscripts reporting work that falls within the International Committee of Medical Journal Editors (ICMJE) definition of a clinical trial: any research study that prospectively assigns human subjects to one or more health related interventions to evaluate the effects on health outcomes. Please note that we only accept registration numbers issued by registries that meet all of the ICMJE criteria (http://www.icmje.org/about-icmje/faqs/clinical-trials-registration/). Registries which meet the requirements of the ICMJE include WHO Primary Registries (http://www.who.int/ictrp/network/primary/en/index.html). This could be done retrospectively since your study is completed yet.

Once you know your trial registration number, please submit a revised version of your manuscript with the number included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number and the date of registration, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Retrospectively registered 28 September 2004. Please note that there should be no space between the letters and numbers of the trial registration number.

Answer: We know our trial registration number (IRB number), so we submit a revised version of our manuscript with the number included in the ‘methods’.
2. Please move keywords on Title page to the end of "Abstract" section.

3. Please move Abbreviations on Title page to the end of main text, following "Conclusion" section, please also arrange a "Conclusion" section following "Discussion".

4. Please move Declarations to the end of main text, following "Abbreviations"

5. Please remove "All authors consent for publication", include a "Consent for publication" subsection, and state "Not applicable".

6. Please kindly provide us a clean final version without track changes.

Answer: We modified all as you said (2~6), thank you.